# A Survey on Computer Vision for Assistive Medical Diagnosis From Faces

Jérôme Thevenot<sup>10</sup>, Miguel Bordallo López<sup>10</sup>, and Abdenour Hadid<sup>10</sup>

Abstract-Automatic medical diagnosis is an emerging center of interest in computer vision as it provides unobtrusive objective information on a patient's condition. The face, as a mirror of health status, can reveal symptomatic indications of specific diseases. Thus, the detection of facial abnormalities or atypical features is at upmost importance when it comes to medical diagnostics. This survey aims to give an overview of the recent developments in medical diagnostics from facial images based on computer vision methods. Various approaches have been considered to assess facial symptoms and to eventually provide further help to the practitioners. However, the developed tools are still seldom used in clinical practice, since their reliability is still a concern due to the lack of clinical validation of the methodologies and their inadequate applicability. Nonetheless, efforts are being made to provide robust solutions suitable for healthcare environments, by dealing with practical issues such as real-time assessment or patients positioning. This survey provides an updated collection of the most relevant and innovative solutions in facial images analysis. The findings show that with the help of computer vision methods, over 30 medical conditions can be preliminarily diagnosed from the automatic detection of some of their symptoms. Furthermore, future perspectives, such as the need for interdisciplinary collaboration and collecting publicly available databases, are highlighted.

*Index Terms*—Computer vision, face analysis, facial symptoms, imaging, medical diagnosis.

### I. INTRODUCTION

T HE global increase in life expectancy within the world population during the past century has been possible as a result of the improved access to clinical facilities and developments in medical diagnostics. To further enhance diagnosis with an objective second assessment, computer-based solutions have recently been developed to help in the early detection of some diseases. In addition, as a result of the expansion of the human lifespan and its quality, the boost of clinical needs induces high costs for the society. These costs are suitable to be reduced with the inclusion of automatic processes. Vision is a key component

Manuscript received April 26, 2017; revised July 28, 2017 and September 13, 2017; accepted September 13, 2017. Date of publication October 4, 2017; date of current version August 31, 2018. (Corresponding author: Jérôme Thevenot.)

J. Thevenot is with the Medical Imaging, Physics and Technology Research Unit and the Center for Machine Vision and Signal Analysis, University of Oulu, Oulu 90014, Finland (e-mail: jerome.thevenot@oulu.fi).

M. B. López and A. Hadid are with the Center for Machine Vision and Signal Analysis, University of Oulu, Oulu 90014, Finland (e-mail: miguelbl@ee.oulu.fi; hadid@ee.oulu.fi).

Digital Object Identifier 10.1109/JBHI.2017.2754861

for building artificial systems that can perceive and understand their environment, similarly to humans who perceive the great majority of information about their environment through sight. During the past decade, there have been numerous research and development efforts in the field of wearable health monitoring systems that were motivated by the need to monitor a person's health status outside of the hospital [1]. However, most proposed techniques for health monitoring require users to attach bulky sensors, chest straps or sticky electrodes. This obviously discourages regular use because the sensors can be uncomfortable or encumbering. To confront these issues, the development of non-contact healthcare has increased, along with technologies robust and simple to use for the diagnostic of conditions and the follow-up of patients [2], [3]. Camera-based methods offer an unobtrusive solution for the monitoring and diagnosis of subjects. Recording devices such as web cameras or smartphones are nowadays common tools, providing an easy access solution of any physiological measurements reachable by such technologies.

The human face houses most of the sensory apparatus – eyes, ears, mouth, and nose - allowing the bearer to see, hear, taste, and smell. Apart from these biological functions, it also provides several signals about health. Indeed, certain medical conditions alter the expression or appearance of the face due to physiological or behavioral responses. These facial signs of disease can provide information to the clinician concerning the state of the patient. From a computer vision viewpoint, detecting abnormalities in patient facial structures and/or expressions is a challenging research problem. The critical issues concern the establishment of basic understanding of the correlations between the face symptoms and the health conditions and to the development of computational models that encode the identified correlations. The main issue is the lack of a code-book, which maps the range of facial patterns to clinical conditions based on proven medical evidences (e.g., indicating that many diseases and brain disorders produce facial abnormalities and interrupt normal facial expression).

Establishing the relationship between digital images and clinical information is a challenging task that can be overcome by detecting medical symptoms. For example, a multitude of genetic syndromes can cause craniofacial abnormalities that can eventually come in pairs with a defect on other organs or systems [4]. While some of these syndromes can have very distinctive facial features, others can be harder to detect at first sight. The study of facial morphology is highly interesting when it concerns the discrimination of craniofacial pathologies,

<sup>2168-2194 © 2017</sup> IEEE. Translations and content mining are permitted for academic research only. Personal use is also permitted, but republication/redistribution requires IEEE permission. See http://www.ieee.org/publications\_standards/publications/rights/index.html for more information.

requiring for this purpose the establishment of robust facial landmarks [5]. Furthermore, considerations on how to define a normal face [6] from a point of view of aesthetic and social interactions have generated multiple studies to characterize facial morphological features such as facial symmetry or skin color. While evaluating craniofacial abnormalities is straightforward by comparing anatomical measurements of a subject with the ones of an average healthy individual, the facial diagnostic of other conditions can be more challenging. Furthermore, clinicians provide a subjective assessment which can vary among the practitioners, and be influenced by the ethnical background of the patient [7]. For this purpose, recent literature has exposed the apparition of new technologies as well as the development of image processing tools to capture the required data, essential for clinical assessment of health conditions in a subjective manner. The main idea behind using computer vision for a clinical purpose would be to decrease the errors related to human factor in the decision-process regarding patients. Furthermore, while medical examinations are costly, analysis based on facial imaging remain low-cost and their developments could eventually lead to a reduction of healthcare expenditures.

The aim of this article is to provide the reader with an overview of the research problem with references to existing works, to stimulate the research and to help unifying the efforts towards the development of new tools and databases for evaluating and monitoring the progress in the field. In this regard, this article can be used as a reference of both experienced and non-experienced researchers. Only distant non-invasive imaging modalities available to the public were chosen, to provide reproducible studies for research groups without access to expensive or encumbering medical equipment such as magnetic resonance imaging or computed tomography. This survey article presents a general overview of different works in the literature on using computer vision for health diagnosis from faces. It also discusses the open issues and future directions. The focus is not only on conventional sensing technologies (e.g., digital imaging) but also on novel imaging methods such as thermography or stereo-photogrammetry. From this survey, the reader will get an understanding of the challenges inherent of the detection of different medical conditions and how they have been tackled so far. Eventually, some leads for new research based on alternative approaches or combination of existing ones could be inspired from this review.

The article is organized as follows. An overview of the different imaging techniques, their particularities and limitations is described in Section II. Section III gives a typical description of the methodological workflow of the facial analysis process. Section IV provides a brief description of the facial anatomy. Section V reports specific medical conditions with computerassisted solutions. This non-exhaustive list attends to present the different options available for an automatic diagnostic approach. The main findings and references are summarized in Section VI. Section VII discusses the significance of the findings, and provides futures perspectives and requirements of computer-based diagnostics of medical conditions from facial imaging, while Section VIII concludes the article.

### II. FACIAL IMAGING MODALITIES

The nature of the imaging data can be diverse depending on relevant facial features to assess and the aim of the study. While conventional options such as basic digital imaging or video have been widely used, some alternative options have emerged to collect further information. The continuous reduction of costs for new imaging devices, as well as the late development of new tools in image processing, have further encouraged the studies of these alternative solutions [8], [9]. This section briefly introduces the possibilities in terms of facial imaging modalities, with a short description of their strengths to detect specific symptomatic clues.

Conventional digital cameras can encode the visible light in three different channels, (red, green and blue). As the most common option for assessment of skin color [10] and facial morphology [11], the ability of conventional digital imaging to discriminate specific health conditions has been extensively studied. However, one drawback of two-dimensional imaging applied to facial morphology is its inaptitude to provide depth of the anatomical landmarks, an issue solved using stereo photogrammetry [12].

The technique of stereo photogrammetry consists in generating a three-dimensional surface from a set of images acquired from different locations. Multiple images are combined through computational models to estimate the actual 3D position of the scanned environment. This approach permits to obtain 3D coordinates of object points from triangulation and reconstructing its volume from them, allowing accurate volumetric distance measurements. Due to its spatial accuracy, this technique has been largely used in studies of craniofacial deformities [13], since an accurate volumetric delimitation of facial edges can be obtained. Following this trend, extensive research has been performed to represent the face volumetrically [14], capturing subtle information of conditions (e.g., schizophrenia) [15]. The recent introduction of low-cost depth cameras (e.g., Microsoft Kinect 1&2) provides exciting new opportunities for detailed facial image analysis. For example, the study of Nakamura et al. [16] used low-cost depth camera to grade torticollis severity from the patient's facial direction and tilt. Furthermore, these sensors have already been applied in clinical applications related to reeducation and therapeutic exercises [17].

Video cameras allow the recording of the temporal variations of a scanned environment, enabling the assessment of small movements or color variations, even when they are unperceivable by the human eye. Videos are necessary for specific conditions related to motion or presenting symptoms solely perceived over a certain period. For example, multiple studies related to eye tracking have been performed to correlate visual behaviors with psychiatric [18] or anxiety disorders [19]. However, while video imaging provides more information than basic digital imaging, the nature of the data remains very similar.

There is an increase of studies using imaging modalities focusing on a different spectral range than the conventional visible one, providing new leads of research. As one of them, thermal imaging is a non-contact, non-invasive imaging method which provides information on human body temperature by assessing the infrared spectrum of the subject. Thermal cameras can detect radiation in the infrared range of the electromagnetic spectrum, usually from 0.9 to 20  $\mu$ m, converting the amount of radiation into visible images. Since infrared radiation is directly proportional to the temperature emitted by all entities, thermal images make it possible to see objects even without visible illumination. The nature of the information provided by this method makes it highly relevant for applications related to clinical medicine [20]. Indeed, not only the thermal-print provides information on the shape of the face [21], but also multiple contributing thermal factors can be assessed (e.g., blood flow, cell metabolism, sweat gland activation), causing local changes in superficial skin temperature. More specifically, different reasons such as inflammatory processes [22], fever [23], [24], cancers [25] or even medications [26] can be responsible for changes in the skin temperature. Due to the increase of thermal camera accuracy and resolution [27], research has been performed lately to evaluate how much information the distribution of heat in the face can provide. For example, various applications related to mass-screening have been using thermal imaging as a fever detection tool [28] for severe acute respiratory syndrome related to H5N1 or more recently to Ebola virus.

## III. COMPUTER VISION METHODOLOGY FOR FACIAL ANALYSIS

The computer vision methodology applied in assistive medical diagnosis typically follows a similar structure: it relies on the extraction and combination of several face descriptors obtained from pre-processed images and uses them to build models that can generalize to unseen data. These features, extracted both in the spatial and spatio-temporal data, are then input into classifiers to predict if the new sample if of a certain type (classification) or to assess the severity of a symptom or condition in a continuous way (regression).

As preprocessing, the first step of most computer vision techniques consists in segmenting specific regions of interest from each sample and detecting either faces or landmarks. The regions containing the useful facial information are cropped and registered into a predefined template, usually preserving the interpupillary distance (used as reference). The selection of useful information is derived from the location of the symptoms to evaluate. For example, studies on facial abnormalities will focus on regions known to be affected and will compare morphological measures with the ones expected in "healthy" subjects. An eventual secondary step in preprocessing is the conversion of the colorspace, either to reduce the complexity of the model by using solely grayscale values if sufficient, or to enhance specific color-based features if necessary (eg. studies related melanoma, hepatitis, etc.).

Local descriptors based on texture analysis such as variants of Histograms of Oriented Gradients (HOG) [29] or Local Binary Patterns (LBP) [30], have also been extensively exploited as they can evaluate local characteristics that can be affected by the medical condition. A typical methodology used with handcrafted features encodes the face information by dividing each frame into blocks and extracting multiscale features from the ones within the regions of interest of the studied disease, concatenating the results into a high dimensional descriptor to exploit multiple information simultaneously. However, since this combination generates very large feature vectors, dimensionality reduction techniques have been extensively used. Among the most used methods are Principal Component Analysis (PCA), Independent component Analysis (ICA), or variations that try to learn the weight of each feature component.

The next step after feature extraction is the classification, with complexity of methods varying based on the discriminative power of the features. Cosine similarity and K-Nearest-Neighbors have been used with mixed performance, while model-based classification utilizing a bi-class support vector machine (SVM), shows to be the preferred method when the amount of data allows the partition into meaningful splits of training and testing data. As the result of the analysis, the postprocessing provides the "final verdict" on the medical condition targeted and can eventually show which features suggested the decision, as a supporting tool for the user.

Most of the existing work is still not associated with the recent significant progress in the machine learning field that suggests the use of deep features and neural networks. Deep learning methods have recently had high successes in medical imaging field [31], [32], but their use in clinical practices still remains limited. The reason behind this may be the scarcity of training data, since these approaches require enough training samples, which are not always available.

## IV. FACE ANATOMY

To consider the face as a mirror of the health, it is important to understand what facial imaging is reflecting. A complex biological system lies below the superficial layer of the skin and each of its component can affect what can be seen. Eventually, the facial appearance depends on several factors. The most important is the morphology of the skull, defined by the concavities and convexities of the bones underlying the soft tissues. Thus, facial aesthetics are highly pre-determined by the facial skeleton features, such as the prominence of cheekbones or the protuberances in the inferior mandible [33].

Besides skull morphology, the appearance of the face is affected by features such as muscles to perform expressions, innervation allowing blood supply (venous drainage and branches of the carotid system), cutaneous sensory innervation [34] or fat compartments. Fig. 1 depicts the superficial fat compartments of the face, contributing to the face aspect. The activity of the facial muscles can reveal important information on the health condition of a subject. For example, while the partial loss of muscle activity is symptomatic of facial paralysis, involuntary movements can be characteristics of multiple conditions such as hemifacial spasm, Meige syndrome, dyskinesia, tics [35]. All the features related to the various systems and their interactions have to be taken into consideration in treatments involving facial procedures. While the injury of a facial nerve can affect its function and lead to clinical deficits, it can also result in local paralysis if a motor nerve is involved, to loss of sensation, decrease of taste, abnormalities in lacrimal activity or even salivary deficiency in cases of other nerves affected [36].



Fig. 1. The superficial fat compartments of the face.

Facial mapping, corresponding to the segmentation of the face into different regions of interest, has been extensively used in skin analysis based on facial location and the properties of the underlying tissues. From a point of view of computer vision, mapping the face allows to focus on areas that are highly relevant to specific conditions. Typically, the areas are defined from distinct facial anatomical landmarks such as the edges of the lips, the eyes, the nose and the contours of the face [37]–[39]. The areas defined between these landmarks have different information inherent of their locations. Eventually, in medical applications, the combination of data collected in each region of interest can provide indications on specific conditions.

## V. AUTOMATIC DIAGNOSIS FROM FACIAL IMAGES

The assessment of facial symptoms can be crucial for the diagnostic of specific medical conditions. While a practitioner can study different facial features from careful observation, in some cases the challenging evaluation of subtle changes and their subjective assessment can alter the reliability of the diagnostic. Thus, computer vision was suggested as a solution to offer an automatic and objective assessment of facial features to help clinicians in their diagnostic [40]. This section addresses various applications of computer-based facial analysis: from basic monitoring of vital signs and assessment of pain, to the evaluation of specific conditions based on asymmetries, expressions, facial movements or facial heat distribution. For each condition, an overview of the different approaches is given with the most relevant outcomes reported.

## A. Monitoring of Vascular Pulse

Plethysmography is the detection and measurement of volumes changes within the body. Its main application is the monitoring of the vital autonomic functions related to respiration rates and cardiac output (i.e., pulse). The study of cardio-vascular pulse waves traveling through the body and the monitoring of blood flow and its velocity, provide indications in the diagnostic of vascular disease [41], but also can help in the monitoring of conditions related to sudden infant death syndrome, sleep apnoea or pulmonary disease [42]. While different methods have been developed in the past to detect pulse [43], alternative non-contact solutions based on photoplethysmography or thermal imaging, able to capture the pulsatile peripheral blood flow, have been promoted. Typically, the pulse waves from a dedicated light source are evaluated by assessing their variations in transmission and reflection on the surface of the skin. However, recent literature has shown that standard regular cameras can perform the pulse measurement in normal ambient light [44].

It has been suggested that the monitoring of physiological data can be used to assess the impact of physical exercise [45]. Furthermore, non-contact physiological monitoring can be applied in multiple domains such as telemedicine and homecare, since low-cost devices such as webcams, and smartphones can already perform the task [46]–[50]. Camera-based photoplethysmographs studies can be divided into two parts: the selection of regions of interest and their analysis. The first step of the process requires motion correction to reduce the impact of motion artifacts within the selected areas of interest. In this context, the positioning of the patient in a way that minimizes the face tracking is important.

The image analysis of the pulse often follows an approach based on color changes, typically using the information of the green channel of the RGB signal. Indeed, the green channel, highly correlated with variations in light, shows to be the most relevant to capture the pulse, while the red channel is useful to perform illumination correction [51]. The variations of the illumination-independent signal are analyzed along time, consolidating the fundamental signal frequency, corresponding to the pulse, and eliminating the possible noise [47], [48], [52]. Other color-based approaches have been performed using the HUE channel [53] or by a conversion to YCbCr (luma, blue chroma and red chroma) color space [54]. One of the main advantage of color-based analysis is the limited amount of facial pixels required to monitor the vascular pulse [42]. An alternative approach for measuring the pulse from facial video is to capture and track the subtle movements of the face, augmenting them using Eulerian video magnification [55]. Subsequent improvements of the system [56], [57] focused on the vertical movements of the head and the utilization of Fourier or Cosine transforms, and Principal Component Analysis (PCA). These methods obtain results with similar accuracy as color-based methods, but with improved robustness to illumination changes.

In most of these studies, the estimation of pulse is strongly correlated with the heart rate assessed from finger sensors or electrocardiograms, presenting 1–2% errors. However, while the acquisition process for monitoring the vascular pulse has been simplified, further challenges related to motion artifacts such as respiratory and head movements still require to be taken into consideration [51], [54]. It has been recently suggested [58] that using two regions of interest could be a solution to evaluate dynamic heart rate by detecting other physiological signs (e.g., eye blinking, yawning). Investigations are also still required to improve the robustness of the methods for different lighting conditions [59] and skin colors, while keeping the time span required to obtain the first confident evaluation of the heart rate as short as possible.

However, while the previous methods had latencies varying from tens of seconds [47], [48], [51] up to few minutes [50],

TABLE I HOMOLOGY OF PAIN EXPRESSION ACROSS THE LIFESPAN

Muscular basis	Response		
Corrugator	Brow lower <sup>1,2,4,5,6</sup> , brow bulge <sup>3</sup>		
Orbicularis oculi	Eye squeeze <sup><math>1,3,4,5,6</math></sup> , cheek raise <sup><math>1,2,4,5</math></sup> , lids tighten <sup><math>1,2,5</math></sup>		
Levator	Nose wrinkle <sup>1,4,5</sup> , lip raiser <sup>1,2,4,5,6</sup> , eye closure <sup>1,2</sup> , nasolabial furrow <sup>3,4</sup>		
Zygomatic	Lip corner pull <sup>1,2,3,5</sup>		
Risorius	Horizontal mouth stretch <sup>4</sup>		
Pterygoid	Vertical mouth stretch <sup>3,4,5</sup>		
Mentalis	Chin quiver <sup>3</sup>		
Nasalis	Flared nostril <sup>4</sup>		

<sup>1</sup>Prkachin (adults) [61]. <sup>2</sup>Prkachin & Solomon (adults) [62]. <sup>3</sup>Grunau & Craig (neonates) [63]. <sup>4</sup>Gilbert *et al.* (children) [64]. <sup>5</sup>Kunz *et al.* (adults/elderly) [65]. <sup>6</sup>Kunz *et al.* (young adults) [66].

[52] depending on the conditions of imaging acquisition, recent developments succeeded to reduce these times to few seconds [49] Finally, a study by Muender *et al.* [59] evaluated the feasibility of gathering heart rate from online videos as a step towards online physiological data collection. From their study, the potential applicability of heart rate assessment by online support requires not only a minimum network bandwidth (>16 mbps), but also good instructions to the participants to reach minimal offset error (<3 bpm).

## B. Pain

Knowing the patient's pain is very relevant for practitioners and other health-care personnel to provide information on the state of the patient, and to suggest some preliminary medication such as painkillers. Extensive research related to automatic pain assessment has been performed. These studies have pinpointed the correlation of facial expressions with the activity of specific muscles, with regards to the age group population studied. A summary of the pain related facial movements adapted from Prkachin [60] can be seen in Table I. The Prkachin and Solomon FACS (Facial Action Coding System) pain scale is currently the only metric defining pain on a frame-by-frame basis [62]. The automatic recognition of pain expression can contribute to the reliability of pain measurements, improving the subjective assessment by practitioners. Pain assessment is at most relevance when the patient cannot express its suffering (ie. young age, mental condition, physical disability).

Some of the available databases rely on the use of facial videos from simulated pain expressions of actors, such as the Acted Facial Expression Database (STOIC) [67]. The database includes videos from 35 actors among which 15 sequences have been identified as expressions of pain. Another database without referential scale is the Infant Classification of Pain Expression (COPE) database, described in the work of Brahnam *et al.* [68]. This database, composed by 206 color images of 25 participants, has been used in multiple studies for the detection of pain from neonates' facial display [69]–[72]. The lack of baseline reference to grade the pain levels can be addressed by inducing a spontaneous painful reaction from external stimuli. The not publicly available Spontaneous Pain Expression Database

[73] which includes videos of expressions of 20 participants undergoing thermal heat stimulation at non-painful and painful intensities.

A more representative approach of induced pain has been described in the work of Lucey et al. [74], in the UNBC McMaster shoulder Pain Expression Archive database. This database contains more than 200 video sequences composed of 48,000 coded frames and the reported pain scores based on both self-reports and measurements from observers. This database brought a clinical baseline to the studies of pain by using a population suffering shoulder pain. Eventually, Ashraf et al. [75] and Lucey et al. [76] developed a method to detect pain using an active appearance model -based features extraction to cluster frames and to assign them a label to train a SVM classifier. Sikka et al. [77] improved the accuracy of pain classification from 68.31-80.99% up to 83.7% by using a multiple instance learning based on multiple segments representation. Such approach considers temporal information and issues related to label ambiguity. Finally, Kharghanian et al. [78] recently applied an unsupervised feature learning approach, raising the pain detection accuracy up to 87.2%.

An issue raised by databases without physiological information is that pain severity is assumed to be correlated solely with the intensity of the stimulus. This hypothesis does not have clear fundamentals, due to the bias induced by the personal subjectivity of pain. As an alternative robust baseline, the BioVid Heat Pain database [79] provides, in addition to facial expression videos, a synchronized collection of physiological data that measure a set of physical responses such as skin conductance level (SCL), electrocardiogram (ECG) and electroencephalogram (EEG). This database was generated to assess how combined facial features, head pose estimation and facial expression can relate an induced pain to physiological data. This publicly available database features videos of 90 participants grouped by age (18 to 65 years old).

The methodology used in facial pain analysis varies across studies. However, local descriptors have typically been used, as they showed statistically significant outcomes to discriminate pain in enfant (90% accuracy) [72], or when related to estimated pain intensity (Pearson correlation 0.5 to 0.6) [80]. The compression of the data utilizing techniques such as PCA and the selection of robust classifiers like Support Vector Machines (SVM) [69]–[71], [75], [76] are also of importance. Other approaches such as tracking distances between face landmarks and evaluating the number of edge points in the nasal root as a detection of wrinkles have also shown their relevance [73]. Finally, Sikka et al. [81] developed an algorithm based on computer vision and machine learning to monitor the pain of children after appendectomy. Their model performed equivalently to the nurse in detecting pain, suggesting the relevance of developing tools to support the clinical personal.

The main challenge in studies related to pain is the establishment of a ground truth with reliable labels. A novel approach to collect data was suggested in the study of Hasan *et al.* [82], where they collected images and self-reported levels of pain by patients using a mobile app. While the use of mobile apps could ease the data collection process, special attention should



Fig. 2. An example of Bell's palsy.

be given to the reliability of data and ethical concerns which might raise from them. The generation of extensive and representative databases is of crucial importance to the realization of reliable and normalized pain studies.

### C. Facial Paralysis

Facial paralysis commonly refers to the weakness of facial muscles, affecting their ability to perform movements and eventually, on a paralysis of the affected area. While facial paralysis can originate from supranuclear or nuclear lesions affecting the central part of the face, most of the facial nerve paralysis are caused by infranuclear lesions and affect solely one side of the face. Facial paralysis can have multiple causes such as Bell's palsy, infection, trauma, tumors, strokes and different syndromes such as Guillain-Barre or Moebius.

Computer-assisted diagnostic has been developed with a focus on the detection of bilateral asymmetries occurring in facial paralysis when subjects are typically requested to perform facial expressions (e.g., smiling). Image processing methods are able to measure the distances between anatomical landmarks and facial features and compare both sides of the face [83]-[89]. Furthermore, motion components can also be evaluated during the process to estimate facial muscular response from voluntary emotion [85], [90]–[93]. The accuracies of detection reported vary between 88.3% for smartphone camera acquisition (Linear Discriminant Analysis + SVM method) [88], up to 96.3% from conventional video cameras (Active Shape Models + LBP method) [89]. Finally, Hontanilla et al. [94] proposed a 3D system called FACIAL CLIMA to analyze facial motion. Their system, similar to stereo photogrammetry, uses three cameras to automatically assess the motions of anatomical landmarks on the face.

Bell's palsy is the most common cause of facial paralysis and has a poorly understood etiology. Fig. 2 depicts the possible symptoms of Bell's palsy, the most common cause of facial paralysis. Specifically, Bell's palsy is diagnosed clinically by a series of forced expressions to evaluate the weakness of specific muscles of the face. It has been suggested that a simple ruler can be enough to assess the severity of Bell's palsy by measuring bilateral anatomical asymmetries during specific muscles movements [95]. These asymmetries are related to nerve damage, commonly assessed by electromyography which can also evaluate its severity [96]. Eventually, Barbosa *et al.* [97] used both iris segmentation and automatically detected key points location to assess asymmetries and diagnose palsy using a hybrid rule-based and machine learning classifier. As an alternative to standard video imaging, 3D measurements generated from space-coding method [98] as well as Kinect V2 [99], [100] have been suggested to detect facial asymmetries by retrieving 3D coordinates of facial landmarks.

Different studies tried to provide an objective solution to quantify the severity of the condition [84], [86], [89], [99], using the House-Brackmann facial nerve grading as a reference scale. However, this grading system has been criticized as it does not score accurately each facial function [101]. To address this issue, a tool called "Electronic Facial Paralysis Assessment" was described by Banks *et al.* [102], and was later validated against multiple grading scales (Spearman rho: 0.72 - 0.90) [103].

Methods developed for the diagnostic of Bell's palsy have also been tested in similar health conditions involving facial nerve dysfunction. In their studies, Linstrom *et al.* [104] and Wu *et al.* [105] used facial motion analysis to assess not only patients with Bell's palsy, but also patients with acoustic neuroma (intracranial tumor), glomus jugulare tumors, facial neuromas, parotid masses, temporal-bone fractures and essential blepharospasm (abnormal contraction of the eyelid).

Eventually, studies of bilateral asymmetries and synkinesis (involuntary muscular movements) demonstrated the ability of computer-assisted methods to accurately detect facial dysfunctions, reducing the impact of the subjective assessments performed clinically [105]. In addition, it was concluded that during a voluntary emotion, patients tend to exaggerate the muscular activity of the non-affected side to move the side affected by the condition. Methods based on analysis of the eyelid motion (blepharokymography) have also been developed to assess facial paralysis during blinking [106]. As an extended research on facial asymmetry using 3D-dynamics scans, Quan *et al.* [107] showed that captured facial dysfunction can also discriminate stroke patients, also suggested by Matuszewski *et al.* [108]. Similar study was performed using Kinect by Breedon *et al.* [109].

## D. Neurologic Conditions and Neurodevelopmental Disorders

Neurological disorders refer to a group of diseases which affect the brain, spine and the nerves connecting them. Several of these conditions present symptoms suitable for computer vision methods. For example, the severity of neurological conditions such as Parkinson's disease can be assessed by tracking motions of the patients [110]. Similarly, video acquisition is clinically used with patients affected by epilepsy to track seizures [111], mainly to assess changes in both mouth and eye features [112], during an episode of abnormal neurological electric activity such as absence seizure [113].

Some birth defects can be discriminated from facial characteristics. For example, facial structural deformities induced by congenital diseases can be representative of specific conditions



Fig. 3. Facial characteristics associated with fetal alcohol syndrome [132].

[114] and eventually used as classification criteria. In this context, computer-assisted tools have tried to classify people that present certain conditions, such as Down's syndrome [115], the Cornelia de Lange syndrome [116], or other syndromes [117] in an automatic way. Despite Down's syndrome having a defined set of facial features, individuals affected by the syndrome will generally present just seven or eight of them [118]. Moreover, the similitudes between siblings can make it challenging to distinguish people affected from non-affected relatives. In computer-assisted diagnosis, Down's syndrome assessment can be performed from a single image. A typical methodology consists in the detection of anatomical landmarks, then on the application of texture analysis using local descriptors, and finally on classifiers to discriminate affected subjects. Numerous works have focused on the automatic detection of subjects with Down syndrome [11], [115], [119]–[122]. The performance of the automatic discrimination methods reaches up to 97% accuracy from groups of infants. However, ongoing research is still being done, mostly to expand databases and increase the number of relevant features as well as classification methods [115].

The fetal alcohol syndrome is a birth defect related to high consumption of alcohol during pregnancy, which alters the normal development of the fetus and eventually leads to both physical and mental defects of the newborn. Fig. 3 shows the facial characteristics of individuals affected by fetal alcohol syndrome. Extended research has been performed to screen fetal alcohol syndrome from image analysis, based on specific facial phenotypes [123], [124]. Most of the research using computer vision to assess this condition has focused on both the measurements of the eyes and lips, as well as the smoothing of the philtrum. It has been shown that the assessment of the condition can be performed from simple facial images [125]. The obtained error rate is 14.3% of false negatives and no false positives [126].

To improve the assessment of the condition of fetal alcohol syndrome, 3D measurements taking into account the depth of the facial features have been performed by stereo photogrammetry [12], [127]–[130]. These studies lead to the development of a reliable device for anthropometric measurements for small infants. Mutsvangwa *et al.* [128] showed that these measurements can provide different accuracy for discriminating subjects affected by the symptoms depending on the age. Thus, the accuracy of the method for 5 years old subjects reaches

95.46%, while the rate decreased to 80.13% for 12 years old subjects. In addition to age grouping, Fang *et al.* [131] demonstrated that grouping patients based on their ethnic background provides better results. They used 3D facial laser scanning for patients and controls from two study sites (Finland and South Africa), with classification accuracies from 80% to over 90% by separating the cohorts.

The combination of anatomical landmarks and textural information allows to obtain a numeric representation of the face with numbers corresponding to well-defined features. It been suggested that multiple syndromic condition could be classified in clinical practice with such method [133], [134], despite overlapping features occurring in studies on facial dismorphism [135]. However, one drawback is the manual intervention for landmark placement [136], [137].

### E. Psychiatric Disorders

Psychiatric illnesses are disorders associated with abnormal behavioral and mental patterns compared to the social norm, causing suffering or disabilities in the daily life of the person affected. The diagnostic of mental conditions can be challenging due to unclear symptoms inducing a subjective assessment of the illness. For example, attention deficit hyperactivity disorder (ADHD) is the most common psychiatric disorder diagnosed among children. However, this syndrome shares multiple symptoms with other mental illnesses such as bipolar disorder [138]. The nature of ADHD remains poorly understood and its diagnostic still controversial [139], an issue quite common amongst different psychiatric conditions. As ADHD is characterized by symptoms such as hyperactivity, impulsivity, inattention, etc., Kinect has been suggested as a modality to assess behaviors of patients under specific tasks [140]. Eventually, Jaiswal et al. [141] created the database KOOMA, containing video and Kinect recordings of 55 subjects (controls, patients with ADHD/autism) listening, reading stories and answering questions. Using facial expression analysis and 3D analysis of behavior, they were able to reach a classification accuracy of 96% for controls vs condition groups (ADHD/autism), these disorders sharing some common symptoms between each other's.

Multiple studies on mental disorders revealed that the analysis of eye movement could greatly improve the diagnostics not only for ADHD [142], [143], [144], but also other psychiatric conditions [145], [146] such as schizophrenia [147]–[150], bipolarity [18], [150], autism [147], [151]–[153] or social phobia [19]. The basis of these studies is that people affected by these conditions will have specific types of ocular movement that could be symptomatic of illnesses or phobias [154]. Another approach by Benfatto *et al.* [155] was developed based on features extraction and SVM to classify children with dyslexia from control ones with 95.6% accuracy.

As the most common mental disorder, depression has been widely studied in order to quantify its severity in different groups of population [156]–[158]. It has been demonstrated that during a period of depression, a person affected is more likely to present facial expression disturbances, due to mood changes [159], [160]. The lack of objectivity in the measurement

techniques to evaluate depression encouraged the development of computer-assisted diagnostics. Typically, studies on automatic detection of depression are based on the analysis of head movement and facial dynamics from stimuli. Usually, the stimuli consist of specifically designed interviews. The facial responses are collected as a video and analyzed to determine special features associated with depression, with accuracies from 76.7% up to 79.0% [161], [162]. In addition to depression, a similar protocol has previously been applied in a study on schizophrenia by Wang et al. [163], demonstrating the inability of schizophrenic patients to express different emotions. Eventually, thermal imaging has been applied to assess facial thermal changes from induced emotion to classify moderate and high schizophrenia patients [164]. This study used MANOVA and SVM classifiers on features from the forehead, nose (representative area for stress analysis [165]) and right cheek, reaching an identification rate of 94.3%.

## F. Mandibular Disorders

Temporomandibular disorder involves pain as well as a dysfunction of the muscles during mastication, inducing a restricted mandibular movement. The diagnosis is complex since the condition can have different etiologies. The use of thermal imaging has shown mixed accuracy depending on the aimed diagnostic. Extended research on temporomandibular disorder has been performed, mostly from the assessment of heat distribution obtained by thermal imaging [166]-[168]. Despite optimistic earlier studies suggesting the ability of thermal imaging to discriminate patients with an unspecific temporomandibular disorder with an accuracy in the range of 85–90% [169]–[172], recent studies focusing on diagnostics of specific conditions have shown mixed accuracies. For example, while skin temperature was proven to be associated with chronic myogenous temporomandibular disorder [173], thermal imaging failed to provide an accurate diagnostic of both this disorder [174], [175] and arthralgia [176] However in these studies, the severity of the participants' conditions was not assessed before the tests, as demonstrated relevant in a following study [177].

It has been suggested that the studies of further forms of thermal analysis are required, not only to fill the lack of a standardized protocol for temperature measurements [174], but also to gain a deeper understanding of thermal patterns [176]. An improvement of the selection of the regions of interest is required, to assess a larger part of the muscles and to determine accurately which areas are related to specific mandibular disorders. Eventually, Haddad *et al.* [178] considered multiple regions of interest directly related to specific muscles. They reported that thermal imaging could be used as a supporting tool for clinical screening myogenous temporomandibular disorders but they also suggested that a stimulation (thermal, mechanical or chemical) could accentuate the results. This suggestion was further confirmed in another study involving thermal imaging and chewing test [179]. A profile thermal image is depicted in Fig. 4.

## G. Other Conditions

The methods described in previous sections are suitable to be applied in other conditions as well. For example, the use of facial imaging has also been applied to evaluate facial abnormalities,



Fig. 4. Digital and thermal images of a subject with simulated inflammation (hot water exposure).

using facial features and classifiers to assess conditions such as acromegaly [180], [181] and Cushing's syndrome [181], [182] or craniofacial deformations associated with specific syndromes [13], [136].

In ophthalmology, video analysis and eye tracking have been suggested to assess conditions related to eye movement such as vertigo [183]. The monitoring of eyelids can be used in case of ptosis [184], as induced by ocular myasthenia gravis. Digital infrared thermal imaging has also been studied to assess orbital inflammation in Graves' ophthalmopathy [26], [185], or from non-specific conditions [186].

The objective assessment of the psychophysiological state of a subject can be made by measuring the arousal response or stress, where thermal imaging has shown to be a suitable technology [165], [187], [188]. Typically, these studies focus on evaluating the skin heat on regions of interest within the face, the periorbital and nasal areas showing to be the most discriminatory features.

Recent efforts have been made to establish the possible correlation between Traditional Chinese Medicine (TCM) -based face inspections with specific clinical conditions. In this context, the study of Hung *et al.* [189] used a TCM device to correlate face color with pulmonary function as a measure of the severity of bronchial asthma. Liu *et al.* [190] and Wang *et al.* [191] used facial chromaticity to identify patients with hepatitis. These studies suggest that establishing the relationship with chromaticity and medical conditions is of vital importance. The focus of these studies consists of two parts: the colorspace and the classifier used. There is a clear need to establish a publicly available database of pictures covering a wide range of different imaging conditions (e.g., lighting parameters) with a sufficient sample size.

## VI. SUMMARY

The analysis of facial imaging can help to automatically detect some symptomatic facial features, in an objective manner. However, in most medical conditions, the final diagnosis of an illness is obtained from the combination of multiple symptoms. Table II provides an overall summary of the different facial symptoms accessible from automatic facial analysis as described in this survey. The possible conditions associated with these facial features are also reported, pointing out the risk of misdiagnosis. The published work related to each specific condition based on the facial feature is provided as reference. Furthermore,

Facial symptoms	Possible conditions	References	Facial symptoms	Possible conditions	References
facial morphology	Down syndrome <sup>1</sup>	[11], [115], [119]–[122]	disturbances in facial	depression <sup>2</sup>	[161], [162]
abnormalities/	schizophrenia <sup>4</sup>	[15]	expressions	ADHD <sup>2, 5</sup>	[141]
characteristics	acromegaly <sup>1</sup>	[180], [181]		autism <sup>2,5</sup>	[141]
	craniofacial deformations <sup>1, 4</sup>	[13], [136]		schizophrenia <sup>2,3</sup>	[163], [164]
	fetal alcohol exposure <sup>1, 4</sup>	[12], [123]–[131], [135], [136]	abnormal heat in orbital area	Graves'	[26], [185]
				ophthalmopathy <sup>3</sup>	
	Cornelia de Lange	[116], [133], [134]		nonspecific condition	[186]
	syndrome <sup>1</sup>			(Ophthalmology) <sup>3</sup>	
	Cushing's syndrome <sup>1</sup>	[181], [182]			
	other syndromes <sup>1, 4</sup>	[117], [133], [134], [136], [137]	asymmetry + synkinesis	facial nerve	[86], [88], [89],
				dysfunction <sup>1, 2</sup>	[104], [105]
facial asymmetry	facial paralysis <sup>1,5</sup>	[83]–[85], [97]–[100], [102]	abnormal eyelid movement	facial paralysis <sup>5</sup>	[106]
	stroke patients <sup>5</sup>	[107]–[109]		ptosis <sup>2</sup>	[184]
abnormal eye	ADHD <sup>2</sup>	[142]–[144]	abnormal iris area	facial paralysis1,5	[97], [98]
movement	dyslexia <sup>2</sup>	[155]	abnormal muscular response	Bell's palsy and	[85], [90]–[94]
	Parkinson <sup>2</sup>	[144]	_	facial paralysis <sup>2,4</sup>	
	autism <sup>2</sup>	[147], [151]–[153]	abnormal skin color	bronchial asthma1	[189]
	schizophrenia <sup>2</sup>	[147]–[150]	yellowish face color	hepatitis <sup>1</sup>	[190], [191]
	bipolarity <sup>2</sup>	[18], [150]	abnormal facial orientation	torticollis <sup>5</sup>	[16]
	social phobia <sup>2</sup>	[19], [154]	heat in mandibular area	mandibular disorder3	[169]–[179]
	vertigo (ophthalmology) <sup>2</sup>	[183]	abnormal head pose	pain <sup>2</sup>	[79]

TABLE II OVERVIEW OF THE DIFFERENT DETECTABLE FACIAL SYMPTOMS AND THEIR ASSOCIATED MEDICAL CONDITIONS

<sup>1</sup>Standard digital imaging. <sup>2</sup>Standard video imaging. <sup>3</sup>Thermal imaging. <sup>4</sup>Stereo photogrammetry imaging. <sup>5</sup>Other imaging modalities.

#### TABLE III

OVERVIEW OF THE DIFFERENT CONDITIONS DETECTABLE, GROUPED BY IMAGING MODALITIES

Standard digital imaging	Standard video imaging	Thermal imaging	Stereo photogrammetry	Other imaging modalities
Neurodevelopmental and neurologic	Vascular pulse	Psychatric	Neurodevelopmental and neurologic	Facial paralysis
8	Psychatric	Schizophrenia	Fetal alcohol exposure	Bell's palsy (blepharokymography),
Down syndrome, Fetal alcohol exposure, Cornelia de Lange	ADHD, Autism, Depression, Schizophrenia, Bipolarity	Ophtalmology	Morphological abnormalities	Facial paralysis (Microsoft Kinect <sup>TM</sup> / Cartesia <sup>TM</sup> ), Stroke related (3D Dynamic Scans / Microsoft Kinect <sup>TM</sup> )
Morphological abnormalities	Ophtalmology	Inflammation in Grave's and other conditions	Cranofacial deformations	Other conditions
Acromegaly, Cranofacial deformations, Cushing's syndrome	Vertigo, Ptosis	Dentistry	Facial paralysis	Torticollis (Microsoft Kinect <sup>TM</sup> )
Facial paralysis	Facial paralysis	Mandibular disorders		
Other conditions	Bell's palsy, Nerve dysfunction	Other conditions		
Hepatitis, Bronchial asthma	Other conditions	Arousal, Stress, Severe acute respiratory syndromes		
Pain	Dyslexia, Absence seizure Pain	Pain		

multiple imaging modalities are available and can be used to detect specific symptoms requiring specific information from the face. Table III provides an overview of the conditions detectable from each imaging modality.

facial heat increase acute respiratory syndromes<sup>3</sup> [23], [24], [18]

## **VII. DISCUSSION**

Healthcare and wellness are common concerns for all populations. However, fulfilling the needs of the population in terms of healthcare services suggests a high cost for the society, derived partially from medical examination tests and practitioners' expenditures. To reduce these expenses, identifying diseases and

managing patients require some changes in the actual clinical protocols. Ideally, these changes would reduce the amount of patient examinations and decrease the time they spend in clinical facilities. While the time spent during the management of medical conditions can be rarely decreased (e.g., treatments), the resources spent to obtain an accurate diagnosis could be reduced with the help of automatic tools. Eventually, the diagnostics could be performed faster and in a non-invasive way avoiding any subjective bias.

pain<sup>1,2,3</sup>

facial muscular contraction

The face is an interesting region of analysis, as it provides indications on multiple conditions and it is the body area the most accessible (unobtrusive). The present survey provided a list of

[55], [68]–[82]

symptoms that could be assessed from facial imaging. Currently, these symptoms are clinically detected either directly by clinicians or by more expensive modalities. This review suggests that all the symptoms presented could be detected or at least assessed objectively using computer vision methods. As shown in Table III, in some cases the diagnostic of a medical condition can be provided by multiple imaging modalities, based either on the assessment of different symptoms of the disease or from alternative approaches specific to the imaging acquisition itself.

For each medical condition, the combination of multiple methods would reach a clinically acceptable accuracy for diagnostic, usable by the clinician as a support for its own decision [192]. The results from the computer vision analysis would not overrule the physician expertise, but be used for extra information as the patient can be affected by another condition sharing similar symptoms. This statement points out one common drawback of most the studies presented here: the symptoms are assessed in controlled settings (patients affected by a condition vs. healthy control group) without considering interpersonal variability. Indeed, in real clinical environment, patients often present symptoms common to multiple diseases. To consider the variability, two separated models should be considered: a general one for patients with unknown condition giving some hints of possible diagnosis, and a person-specific one obtained from data acquired across time per single patient.

While most of the studies presented here aim to provide further assistance to clinicians in their diagnostic by quantifying specific features from images and videos, they can also be applied to detect a larger range of facial symptomatic characteristics subjectively. The combination of symptoms provides hints to the practitioner to formulate the final diagnostic. This standpoint further supports the need to perform multiple analysis to discriminate a specific condition out of a range of eventual diseases. For this purpose, information easily obtainable such as patient history or anthropometric data could be added in the analysis to help in the automatic diagnostic. Despite the clinical evidences of the studies described here, computer vision methods are still at the research stage in the medical field. One of the challenges of facial diagnostic is the impact of variable environmental conditions involved in long-range image capturing. While some successful applications (e.g., melanoma detection in dermoscopy field, pulse monitoring from light changes in the finger) have shown their applicability in close range image analysis, the variability of the results when capturing the image from long range has not yet been evaluated properly.

The main limitation to the development of computer vision methods for diagnosis from the face is the lack of publicly available data. While deep learning methods would be a good solution to improve the diagnostic tools based on facial features, they require an extensive amount of data to properly train the features extraction and classification models [31], [32]. Furthermore, the lack of transparency of the selected features could be an issue to be addressed for clinical applicability. In this review, the methods have been typically applied to a limited number of databases, reducing the relevancy of the studies. Despite the previously reported benefits of creating databases [193], [194], only a limited amount is available. A reason for this is ethical: the

access to personal medical data is under medical confidentiality agreements and involves ethical permits to perform specific research studies [195]. Thus, spreading the medical data for studies that were not mentioned in the original ethical permit is not allowed. Specifically, in facial analysis, this issue is at upmost importance, since the patient can be identified from its appearance, even when its anonymity is preserved.

To increase the relevance of the studies on facial analysis related to medical conditions, the improvement of ground truth data is required and for this purpose, some solutions should be considered. The development of collaborations between research groups from medical and computer science fields could highly improve the access to patient data. In addition, the integration of several acquisition modalities during patient examination could greatly increase the robustness of the decision systems (e.g., combination of image and clinical data). However, the multidisciplinary nature of the collaborations requires significant coordination efforts and joint facilities.

In perspective, alternative imaging modalities can be considered for computer vision methods for clinical purpose. The first one is mobile phones, the constant improvement of the hardware and the quality of their embedded cameras make them a perfect candidate for self-assessment. While multiple mobile applications related to health are available (e.g., Facial Dysmorphology Analysis technology Face2Gene [116], [117]), their use for clinical diagnostic remains limited despite being technically possible [88]. Furthermore, new hardware are being released (e.g., FLIR One), allowing to mount thermal camera on mobile phones, opening a new range of possibilities. Face images acquired using conventional cameras may indeed have inherent restrictions that hinder the inference of some specific details in the face. A promising approach for dealing with those limitations is using images acquired beyond the visible spectrum and/or using non-conventional imaging. With this aim, hyperspectral cameras are able to capture data into multiple bands of the electromagnetic spectrum both in the visible and invisible ranges. This imaging modality allows to identify the molecular composition of the scanned environment and the changes of its spectrum over time in the case of abnormalities due to medical conditions [196]. While hyperspectral imaging studies have mainly focused on getting tissue characterization from smallscale tissue samples [197], new methods for face analysis [198] and tongue cancer diagnosis [199] are emerging.

The combination of multiple imaging modalities and methods could lead to an increase of the accuracy of these predictive tools. However, the lack of standardization of the methodologies due to restricted access to clinical data to validate them has been restraining the emergence of novel solutions to assist the medical community.

### VIII. CONCLUSION

This survey was carried out to assess the current state of evidence related to the implementation of computer vision systems for facial analysis applied to the diagnostic of medical conditions. In this review, a range of distant and non-invasive imaging solutions has been described, providing insights into the suitability of each one of the techniques in the assessment of different types of symptoms related medical conditions.

The analysis of the work related to the traditional approaches of diagnosis from facial observations showed that the establishment of a correlation between face features and clinical data is of high importance. This relationship has been extracted from the review of more than 150 references, where the most relevant methodologies have been identified. The findings show that utilizing computer vision methods, over 30 conditions can be preliminary diagnosed from the automatic detection of some of their symptoms. This review reports clear evidences that the methods presented here could provide valuable tools for practitioners, eliminating subjective bias and reducing diagnostics time and costs. However, these systems still require further validations by clinical trials.

### REFERENCES

- A. Pantelopoulos and N. G. Bourbakis, "A survey on wearable sensorbased systems for health monitoring and prognosis," *IEEE Trans. Syst.*, *Man, Cybern. C, Appl. Rev.*, vol. 40, no. 1, pp. 1–12, Jan. 2010.
- [2] C. Li et al., "A review on recent advances in Doppler radar sensors for noncontact healthcare monitoring," *IEEE Trans. Microw. Theory Techn.*, vol. 61, no. 5, pp. 2046–2060, May 2013.
- [3] A. Mihailidis, B. Carmichael, and J. Boger, "The use of computer vision in an intelligent environment to support aging-in-place, safety, and independence in the home," *IEEE Trans. Inf. Technol. Biomed.*, vol. 8, no. 3, pp. 238–247, Sep. 2004.
- [4] R. M. Winter, "What's in a face?," *Nature Genet.*, vol. 12, pp. 124–129, 1996.
- [5] T. S. Douglas, "Image processing for craniofacial landmark identification and measurement: A review of photogrammetry and cephalometry," *Comput. Med. Imag. Graph.*, vol. 28, no. 7, pp. 401–409, 2004.
- [6] D. Y. Tsao and W. A. Freiwald, "What's so special about the average face?," *Trends Cogn. Sci.*, vol. 10, no. 9, pp. 391–393, 2006.
- [7] A. Lumaka *et al.*, "Facial dysmorphism is influenced by ethnic background of the patient and of the evaluator," *Clin. Genet.*, vol. 92, no. 2, pp. 166–171, 2017.
- [8] E. F. J. Ring and K. Ammer, "Infrared thermal imaging in medicine," *Physiol. Meas.*, vol. 33, pp. R33–R46, 2012.
- [9] G. Lu and B. Fei, "Medical hyperspectral imaging: A review," J. Biomed. Opt., vol. 19, pp. 1–24, 2014, Art. no. 010901.
- [10] S. Zheng *et al.*, "Objective analysis of face feature and expression in traditional Chinese medicine," in *Proc. 5th Int. Conf. Bioinformat. Biomed. Eng.*, 2011, pp. 1–4.
- [11] Q. Zhao, N. Werghi, K. Okada, K. Rosenbaum, M. Summar, and L. M. Zhao, "Ensemble learning for the detection of facial dysmorphology," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2014, vol. 36, pp. 754–757.
- [12] T. Mutsvangwa and T. S. Douglas, "Morphometric analysis of facial landmark data to characterize the facial phenotype associated with fetal alcohol syndrome," *J Anatomy*, vol. 210, no. 2, pp. 209–220, 2007.
- [13] P. R. Ladeira, E. O. Bastos, J. V. Vanini, and N. Alonso, "Use of stereophotogrammetry for evaluating craniofacial deformities: A systematic review," *Revista Brasileira de Cirurgia Plástica*, vol. 28, no. 1, pp. 147–55, 2013.
- [14] H. Popat *et al.*, "Quantitative analysis of facial movement—A review of three-dimensional imaging techniques," *Comput. Med. Imag. Graph.*, vol. 33, pp. 377–383, 2009.
- [15] R. J. Hennessy, P. A. Baldwin, D. J. Browne, A. Kinsella, and J. L. Waddington, "Three-dimensional laser surface imaging and geometric morphometrics resolve frontonasal dysmorphology in schizophrenia," *Biol. Psychiatry*, vol. 61, no. 10, pp. 1187–1194, 2007.
- [16] T. Nakamura, M. Sato, and H. Kajimoto, "Semi-automatic scoring method for torticollis by using Kinect," *in Proc. 17th Int. Congr. Parkinson's Dis. Movement Disorders*, 2013, vol. 28, suppl. 1, Art. no. 328.
- [17] C. Lanz et al., "Automated classification of therapeutic face exercises using the kinect," in Proc. Conf. Comput. Vis., Imag. Comput. Graph. Theory Appl., 2013, pp. 556–565.
- [18] G. Akinci, E. Polat, and O. M. Kocak, "Neuropsychiatric disorders classification using a video based pupil detection system," in *Proc. Int. Symp. Innov. Intell. Syst. Appl.*, 2012, pp. 1–5.

- [19] H. Grillon *et al.*, "Eye-tracking as diagnosis and assessment tool for social phobia," in *Proc. Virtual Rehabil.*, 2007, pp. 138–145.
- [20] L. J. Jiang et al., "A perspective on medical infrared imaging," J. Med. Eng. Technol., vol. 29, no. 6, pp. 257–67, 2005.
- [21] Y. F. Zhao et al., "Automatic location of facial acupuncture-point based on content of infrared thermal image," in Proc. 5th Int. Conf. Comput. Sci. Educ., 2010, pp. 65–68.
- [22] G. Varju, C. F. Pieper, J. B. Renner, and V. B. Kraus, "Assessment of hand osteoarthritis: Correlation between thermographic and radiographic methods," *Rheumatology*, vol. 43, no. 7, pp. 915–919, 2004.
- [23] W. T. Chiu *et al.*, "Infrared thermography to mass-screen suspected SARS patients with fever," *Asia Pac. J. Public Health*, vol. 17, no. 1, pp. 26–28, 2005.
- [24] M. F. Chiang *et al.*, "Mass screening of suspected febrile patients with remote-sensing infrared thermography: Alarm temperature and optimal distance," *J. Formos Med. Assoc.*, vol. 107, no. 12, pp. 937–44, 2008.
- [25] H. Madhu et al., "Extraction of medically interpretable features for classification of malignancy in breast thermography," in Proc. IEEE Annu. Int. Conf. Eng. Med. Biol. Soc., 2016, pp. 1062–1065.
- [26] T. C. Chang, Y. L Hsiao, and S. L. Liao, "Application of digital infrared thermal imaging in determining inflammatory state and follow-up effect of methylprednisolone pulse therapy in patients with graves' ophthalmopathy," *Graefes Arch. Clin. Exp. Ophthalmol.*, vol. 246, no. 1, pp. 45–49, 2008.
- [27] E. F. Ring and K. Ammer, "Infrared thermal imaging in medicine," *Physiol. Meas.*, vol. 33, no. 3, pp. 33–46, 2012.
- [28] E. Y. Ng and R. U. Acharya, "Remote-sensing infrared thermography," *IEEE Eng. Med. Biol. Mag.*, vol. 28, no. 1, pp. 76–83, Jan./Feb. 2009.
- [29] N. Dalal and B. Triggs, "Histograms of oriented gradients for human detection," in *Proc. IEEE Comput. Soc. Conf. Comput. Vis. Pattern Recognit.*, 2005, vol. 1, pp. 886–893.
- [30] T. Ahonen, A. Hadid, and M. Pietikainen, "Face description with local binary patterns: Application to face recognition," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 28, no. 12, pp. 2037–2041, Dec. 2006.
- [31] G. Litjens et al., "A survey on deep learning in medical image analysis," Med Image Anal., vol. 42, pp. 60–88, 2017.
- [32] D. Shen, G. Wu, and H. I. Suk, "Deep learning in medical image analysis," *Annu. Rev. Biomed. Eng.*, vol. 19, pp. 221–248, 2017.
- [33] P. Prendergast, "Anatomy of the face and neck," in *Cosmetic Surgery— Art and Techniques*. New York, NY, USA: Springer, 2012, ch. 2.
- [34] B. Bentsianov and A. Blitzer, "Facial anatomy," *Clin. Dermatol.*, vol. 22, pp. 3–13, 2004.
- [35] N. C. Tan, L. L. Chan, and E. K. Tan, "Hemifacial spasm and involuntary facial movements," *Quart. J. Med.*, vol. 95, pp. 493–500, 2002.
- [36] T. M. Myckatyn and S. E. Mackinnon, "A review of research endeavors to optimize peripheral nerve reconstruction," *Neurol. Res.*, vol. 26, no. 2, pp. 124–138, 2004.
- [37] L. Chevallier, J. R. Vigouroux, A. Goguey, and A. Ozerov, "Facial landmarks localization estimation by cascaded boosted regression," in *Proc. Int. Conf. Comput. Vis. Theory Appl.*, 2013, pp. 103–115.
- [38] F. Zhou, J. Brandt, and Z. Lin, "Exemplar-based graph matching for robust facial landmark localization," in *Proc. Int. Conf. Comput. Vis.*, 2013, pp. 1025–1032.
- [39] Z. Zhang, P. Luo, C. C. Loy, and X. Tang, "Facial landmark detection by deep multi-task learning," in *Proc. Eur. Conf. Comput. Vis.*, 2014, pp. 94–108.
- [40] K. Doi, "Computer-aided diagnosis in medical imaging: Historical review, current status and future potential," *Comput. Med. Imag. Graph.*, vol. 31, no. 4, pp. 198–211, 2007.
- [41] J. Allen, "Photoplethysmography and its application in clinical physiological measurement," *Physiol. Meas.*, vol. 28, no. 3, pp. R31–R39, 2007.
- [42] K. Mannapperuma *et al.*, "Performance limits of ICA-based heart rate identification techniques in imaging photoplethysmography," *Physiol. Meas.*, vol. 36, no. 1, pp. 67–83, 2015.
- [43] A. A. Graham, "Plethysmography: Safety, effectiveness, and clinical utility in diagnosing vascular disease," *Health Technol. Assess*, vol. 7, pp. 1–46, 1996.
- [44] M. Z. Poh, D. McDu\_, J. Picard, and W. Rosalind, "Non-contact, automated cardiac pulse measurements using video imaging and blind source separation," *Opt. Express*, vol. 18, no. 10, pp. 10762–10774, 2010.
- [45] Y. Sun *et al.*, "Motion-compensated noncontact imaging photoplethysmography to monitor cardiorespiratory status during exercise," *J. Biomed. Opt.*, vol. 16, no. 7, pp. 1–9., 2011, Art. no. 077010.

- [46] Y. Sun *et al.*, "Use of ambient light in remote photoplethysmographic systems: Comparison between a high-performance camera and a lowcost webcam," *J. Biomed. Opt.*, vol. 17, no. 3, 2012, Art. no. 037005, pp. 1–10.
- [47] W. Verkruysse, L. Svaasand, and J. S. Nelson, "Remote plethysmographic imaging using ambient light," *Opt. Express*, vol. 16, no. 26, pp. 21434–21445, 2008.
- [48] K. Sungjun, K. Hyunseok, and S. P. Kwang, "Validation of heart rate extraction using video imaging on a built-in camera system of a smartphone," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2012, pp. 2174–177.
- [49] J. Lin, D. Rozado, and A. Duenser, "Improving video based heart rate monitoring," *Stud. Health Technol. Informat.*, vol. 214, pp. 146–151, 2015.
- [50] M. Lewandowska, J. Rumiński, T. Kocejko, and J. Nowak, "Measuring pulse rate with a webcam—A non-contact method for evaluating cardiac activity," in *Proc. Conf. Human Factors Comput. Syst.*, 2016, pp. 4562–4567.
- [51] L. A. Aarts *et al.*, "Non-contact heart rate monitoring utilizing camera photoplethysmography in the neonatal intensive care unit—A pilot study," *Early Human Dev.*, vol. 89, no. 12, pp. 943–948, 2013.
- [52] M. Z. Poh *et al.*, "Advancements in noncontact, multiparameter physiological measurements using a webcam," *IEEE Trans. Biomed. Eng.*, vol. 58, pp. 7–11, Jan. 2011.
- [53] G. R. Tsouri and Z. Li, "On the benefits of alternative color spaces for noncontact heart rate measurements using standard red-green-blue cameras," J. Biomed. Opt., vol. 20, no. 4, pp. 1–6, 2015, Art. no. 048002.
- [54] F. Bousefsaf, C. Maaoui, and A. Pruski, "Continuous wavelet filtering on webcam photoplethysmographic signals to remotely assess the instantaneous heart rate," *Biomed. Signal Process. Control*, vol. 8, no. 6, pp. 568–574, 2013.
- [55] H. Y.Wu, M. Rubinstein, E. Shih, J. Guttag, F. Durand, and W. Freeman, "Eulerian video magnification for revealing subtle changes in the world," *ACM Trans. Graph.*, vol. 31, no. 4, 2012, Art. no. 65
- [56] G. Balakrishnan, F. Durand, and J. Guttag, "Detecting pulse from head motions in video," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2013, pp. 3430–3437.
- [57] R. Irani, K. Nasrollahi, and T. B. Moeslund, "Improved pulse detection from head motions using DCT," in *Proc. Conf. Comput. Vis., Imag. Comput. Graph. Theory Appl.*, 2014, pp. 118–124.
- [58] B. Wei, X. He, C. Zhang, and X. Wu, "Non-contact, synchronous dynamic measurement of respiratory rate and heart rate based on dual sensitive regions," *Biomed. Eng. Online*, vol. 16, pp. 1–21, 2017, Art. no. 17.
- [59] T. Muender, M. K. Miller, M. V. Birk, and R. L. Mandryka, "Extracting heart rate from videos of online participants," in *Proc. Conf. Human Factors Comput. Syst.*, 2016, pp. 4562–4567.
- [60] K. M. Prkachin, "Assessing pain by facial expression: Facial expression as nexus," *Pain Res. Manag.*, vol. 14, no. 1, pp. 53–58, 2009.
- [61] K. M. Prkachin, "The consistency of facial expressions of pain: A comparison across modalities," *Pain*, vol. 51, pp. 297–306, 1992.
- [62] K. M. Prkachin and P. E. Solomon, "The structure, reliability and validity of pain expression: Evidence from patients with shoulder pain," *Pain*, vol. 139, pp. 267–274, 2008.
- [63] R. V. Grunau and K. D. Craig, "Pain expression in neonates: Facial action and cry," *Pain*, vol. 28, pp. 395–410, 1987.
- [64] C. A. Gilbert *et al.*, "Postoperative pain expression in preschool children: Validation of the child facial coding system," *Clin. J. Pain*, vol. 15, no. 3 pp. 192–200, 1999.
- [65] M. Kunz, V. Mylius, K. Schepelmann, and S. Lautenbacher, "Impact of age on the facial expression of pain," *J. Psychosomatic Res.*, vol. 64, no. 3, pp. 311–318, 2008.
- [66] M. Kunz, J. Peter, S. Huster, and S. Lautenbacher, "Pain and disgust: The facial signaling of two aversive bodily experiences," *PLoS One*, vol. 8, no. 12, 2013, Art. no. e83277.
- [67] S. Roy, C. Roy, I. Fortin, C. Either-Majcher, P. Belin, and F. Gosselin, "A dynamic facial expression database," J. Vis., vol. 7, no. 9, p. 944, Jun. 2007.
- [68] S. Brahnam, L. Nanni, and R. Sexton, "Introduction to neonatal facial pain detection using common and advanced face classification techniques," *Stud. Comput. Intell.*, vol. 48, pp. 225–253, 2007.
- [69] S. Brahnam, C. F. Chuang, F. Y. Shih, and M. R. Slack, "Machine recognition and representation of neonatal facial displays of acute pain," *Artif. Intell. Med.*, vol. 36, no. 3, pp. 211–222, 2006.

- [70] S. Brahnam et al., "SVM classification of neonatal facial images of pain," in *International Workshop on Fuzzy Logic and Applications* (Volume 3849 of Lecture Notes in Computer Science). New York, NY, USA: Springer, 2006, pp. 121–128.
- [71] B. Gholami, W. M. Haddad, and A. R. Tannenbaum, "Agitation and pain assessment using digital imaging," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2009, pp. 2176–2179.
- [72] M. N. Mansor and M. N. Rejab, "A computational model of the infant pain impressions with Gaussian and nearest mean classifier," in *Proc. IEEE Int. Conf. Control Syst., Comput. Eng.*, 2013, pp. 249–253.
- [73] Z. Hammal, M. Kunz, M. Arguin, and F. Gosselin, "Spontaneous pain expression recognition in video sequences," in *Proc. Int. Conf. Vis. Comput. Sci.*, 2008, pp. 191–210.
- [74] P. Lucey, J. F. Cohn, K. M. Prkachin, and I. Matthews, "Painful data: The UNBC-McMaster shoulder pain expression archive database," in *Proc. Int. Conf. Automat. Face Gesture Recognit. Workshops*, 2011, pp. 57–64.
- [75] A. Ashraf *et al.*, "The painful face-pain expression recognition using active appearance models," *Image Vis. Comput.*, vol. 27, no. 12, pp 1788–1796, 2009.
- [76] P. Lucey, J. Cohn, S. Lucey, I. Matthews, S. Sridharan, and K. M. Prkachin, "Automatically detecting pain using facial actions," in *Proc. Int. Conf. Affective Comput. Intell. Interact. Workshops*, 2009, pp. 1–8.
- [77] K. Sikka, A. Dhall, and M. Stewart Bartlett, "Classification and weakly supervised pain localization using multiple segment representation," *Im*age Vis. Comput., vol. 32, no. 10, pp. 659–670, 2014.
- [78] R. Kharghanian, A. Peiravi, and F. Moradi, "Pain detection from facial image using unsupervised feature learning approach," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2016, pp. 419–422.
- [79] P. Werner, A. Al-Hamadi, R. Niese, S. Walter, S. Gruss, and H. C. Traue, "Towards pain monitoring: Facial expression, head pose, a new database, an automatic system and remaining challenges," in *Proc. Brit. Mach. Vis. Conf.*, 2013, Art. no. 119, pp. 1–13.
- [80] S. Kaltwang, O. Rudovic, and M. Pantic, "Continuous pain intensity estimation from facial expressions," in *Advances in Visual Computing: ISVC 2012* (vol. 7432 Lecture Notes in Computer Science). 2012, pp. 368–377.
- [81] K. Sikka *et al.*, "Automated assessment of children's postoperative pain using computer vision," *Pediatrics*, vol. 136, no. 1, pp 124–131, 2015.
- [82] M. K. Hasan et al., "Pain level detection from facial image captured by smartphone," J. Inf. Process., vol. 24, no. 4, pp. 598–608, 2016.
- [83] S.Wang and F. Qi, "Computer aided diagnosis of facial paralysis based on Pface," in *Proc. 27th Annu. Conf. IEEE Eng. Med. Biol.*, 2005, pp. 4353–4356.
- [84] I. Song et al., "Profiling bell's palsy based on house-brackmann score," in Proc. IEEE Symp. Comput. Intell. Healthcare E-health, 2013, pp. 1–6.
- [85] T. Tanaka, J. Nemoto, M. Ohta, and T. Kunihiro, "Assessment method of facial palsy by amount of feature point movements at facial expressions," *IEEJ Trans. Electron. Inf. Syst.*, vol. 126, no. 3, pp. 368–375, 2006.
- [86] W. S. Samsudin, K. Sundaraj, A. Ahmad, and H. Salleh, "Initial assessment of facial nerve paralysis based on motion analysis using an optical flow method," *Technol. Health Care*, vol. 24, no. 2, pp. 287–294, 2016.
- [87] K. Ziahosseini, C. Nduka, and R. Malhotra, "Ophthalmic grading of facial paralysis: Need for a closer look," *Brit. J. Ophthalmol.*, vol. 99, no. 9, pp. 1171–1175, 2015.
- [88] H. S. Kim, S. Y. Kim, Y. H. Kim, and K. S. Park, "A smartphone-based automatic diagnosis system for facial nerve palsy," *Sensors*, vol. 15, no. 10, pp. 26756–26768, 2015.
- [89] T. Wang, S. Zhang, J. Dong, L. Liu and H. Yu, "Automatic evaluation of the degree of facial nerve paralysis," *Multimedia Tools Appl.*, vol. 75, no. 19, pp. 11893–11908, 2016.
- [90] T. Tanaka et al., "The evaluation of facial palsy by amount of feature point movements at facial expressions," in Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc., vol. 2, pp. 1463–1466, 2004.
- [91] H. Minamitani *et al.*, "Computerized diagnosis of facial nerve palsy based on optical flow analysis of facial expressions," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2005, pp. 663–666.
- [92] S. He, J. Soraghan, and B. O'Reilly, "Biomedical image sequence analysis with application to automatic quantitative assessment of facial paralysis," *J. Image Video Process.*, vol. 2007, 2007, Art. no. 2.
- [93] S. He, J. Soraghan, and B. O'Reilly, "Objective grading of facial paralysis using local binary patterns in video processing," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2008, pp. 4805–4808.

- [94] B. Hontanilla and C. Auba, "Automatic three-dimensional quantitative analysis for evaluation of facial movement," *J. Plastic, Reconstructive Aesthetic Surg.*, vol. 61, pp. 18–30, 2008.
- [95] R. T. Manktelow, R. M. Zuker, and L. R. Tomat, "Facial paralysis measurement with a handheld ruler," *Plastic Reconstructive Surg.*, vol. 121, no. 2, pp. 435–442, 2008.
- [96] A. A. Pourmomeny and S. Asadi, "Management of synkinesis and asymmetry in facial nerve palsy: A review article," *Iran J. Otorhinolaryngol.*, vol. 26, no. 77, pp. 251–6, 2014.
- [97] J. Barbosa *et al.*, "Efficient quantitative assessment of facial paralysis using iris segmentation and active contour-based key points detection with hybrid classifier," *BMC Med. Imag.*, vol. 16, pp. 1–18, 2016, Art. no. 23.
- [98] S. Katsumi, S. Esaki, K. Hattori, K. Yamano, T. Umezaki, and S. Murakami, "Quantitative analysis of facial palsy using a three-dimensional facial motion measurement system," *Auris Nasus Larynx*, vol. 42, no. 4, pp. 275–283, 2015.
- [99] A. Gaber, M. F. Taher, and M. A. Wahed, "Quantifying facial paralysis using the Kinect v2," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2015, pp. 2497–2501.
- [100] A. Gaber, M. F. Taher and M. A. Wahed, "Automated grading of facial paralysis using the kinect v2: A proof of concept study," in *Proc. Int. Conf. Virtual Rehabil.*, 2015, pp. 258–265.
- [101] T. L. Yen, C. L. Driscoll, and A. K. Lalwani, "Significance of House-Brackmann facial nerve grading global score in the setting of differential facial nerve function," *Otol. Neurotol.*, vol. 24, no. 1, pp. 118–122, 2003.
- [102] C. A. Banks, P. K. Bhama, J. Park, C. R. Hadlock, and T. A. Hadlock, "Clinician-graded electronic facial paralysis assessment: The eFACE," *Plastic Reconstructive Surg.*, vol. 136, no. 2, pp. 223–230, 2015.
- [103] L. S. H Chong *et al.*, "Validation of the clinician-graded electronic facial paralysis assessment," *Plastic Reconstructive Surg.*, vol. 140, no. 1, pp. 159–167, 2017.
- [104] C. J. Linstrom, "Objective facial motion analysis in patients with facial nerve dysfunction," *Laryngoscope*, vol. 112, no. 7, pp. 1129–1147, 2002.
- [105] Z. B. Wu, C. A. Silverman, C. J. Linstrom, B. Tessema, and M. K. Cosetti, "Objective computerized versus subjective analysis of facial synkinesis," *Laryngoscope*, vol. 115, no. 12, pp. 2118–2122, 2005.
- [106] S. H. Choi, T. H. Yoon, K. S. Lee, J. H. Ahn, and J. W. Chung, "Blepharokymographic analysis of eyelid motion in bell's palsy," *Laryngo-scope*, vol. 117, no. 2, pp. 308–312, 2007.
- [107] W. Quan, B. J. Matuszewski, and L. Shark, "Facial asymmetry analysis based on 3-D dynamic scans," in *Proc. IEEE Int. Conf. Syst., Man, Cybern.*, 2012, pp. 2676–2681.
- [108] B. J. Matuszewski et al., "Hi4D-ADSIP 3D dynamic facial articulation database," *Image Vis. Comput.*, vol. 30, no. 10, pp. 713–727, 2012.
- [109] P. Breedon *et al.*, "First for stroke rehabilitation tool: Using the Microsoft 'Kinect' as a facial paralysis stroke," *Int. J. Integr. Care*, vol. 14, no. 8, pp. 1–2, 2014.
- [110] K. Criss and J. McNames, "Video assessment of finger tapping for Parkinson's disease and other movement disorders," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2011, pp. 7123–7126.
- [111] M. Pediaditis *et al.*, "Vision-based motion detection, analysis and recognition of epileptic seizures—A systematic review," *Comput. Methods Programs Biomed.*, vol. 108, no. 3, pp. 1133–1148, 2012.
- [112] M. Pediaditis, M. Tsiknakis, V. Bologna, and P. Vorgia, "Model-free vision-based facial motion analysis in epilepsy," in *Proc. 10th Int. Work-shop Biomed. Eng.*, 2011, vol. 10, pp. 1–4.
- [113] M. Pediaditis *et al.*, "Vision-based absence seizure detection," in *Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2012, pp. 65–68.
- [114] A. Koch and S. Eisig, "Syndromes with unusual facies," Atlas. Oral Maxillofacial Surgery Clin., vol. 22, no. 2, pp. 205–210, 2014.
- [115] Ş. Saraydemir, N. Taşpınar, O. Eroğul, H. Kayserili, and N. Dinçkan, "Down syndrome diagnosis based on Gabor wavelet transform," *J. Med. Syst.*, vol. 3, no. 5, pp. 3205–3213, 2012.
- [116] L. Basel-Vanagaite *et al.*, "Recognition of the Cornelia de Lange syndrome phenotype with facial dysmorphology novel analysis," *Clin. Genet.*, vol. 89, no. 5, pp. 557–563, 2016.
- [117] T. Liehr et al., "Next generation phenotyping in Emanuel and Pallister Killian syndrome using computer-aided facial dysmorphology analysis of 2D photos," *Clin. Genet.*, 2017, doi: 10.1111/cge.13087. [Epub ahead of print].
- [118] "Down's syndrome Scotland," 2009. [Online]. Available: http://www. dsscotland.org.uk/

- [119] K. Burcin and N. V. Vasif, "Down syndrome recognition using local binary patterns and statistical evaluation of the system," *Expert Syst. Appl.*, vol. 38, no. 7, pp. 8690–8695, 2011.
- [120] Q. Zhao et al., "Automated down syndrome detection using facial photographs," in Proc. Annu. Int. Conf. IEEE Eng. Med. Biol. Soc., 2013, pp. 3670–3673.
- [121] Q. Zhao et al., "Hierarchical constrained local model using ICA and its application to down syndrome detection," *Med. Image Comput. Comput.-Assist. Intervention*, vol. 16, no. 2, pp. 222–229, 2013.
- [122] Q. Zhao et al., "Digital facial dysmorphology for genetic screening: Hierarchical constrained local model using ICA," Med. Image Anal., vol. 18, no. 5, pp. 699–710, 2014.
- [123] E. S. Moore *et al.*, "Unique facial features distinguish fetal alcohol syndrome patients and controls in diverse ethnic populations," *Alcohol Clin. Exp. Res.*, vol. 31, no. 10, pp. 1707–1713, 2007.
- [124] T. S. Douglas and T. Mutsvangwa, "A review of facial image analysis for delineation of the facial phenotype associated with fetal alcohol syndrome," *Amer. J. Med. Genet. A*, vol. 152, pp. 528–536, 2010.
- [125] S. J. Astley and S. K. Clarren, "Measuring the facial phenotype of individuals with prenatal alcohol exposure: Correlations with brain dysfunction," *Alcohol Alcoholism*, vol. 36, no. 2, pp. 147–159, 2001.
- [126] S. J. Astley, J. Stachowiak, S. K. Clarren, and C. Clausen, "Application of the fetal alcohol syndrome facial photographic screening tool in a foster care population," *J. Pediatrics*, vol. 141, no. 5, pp. 712–717, 2002.
- [127] T. S. Douglas *et al.*, "Comparison of eye and lip measurements obtained using single and stereophotogrammetry," in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, vol. 1, pp. 545–547, 2003.
- [128] T. Mutsvangwa *et al.*, "Design, construction, and testing of a stereophotogrammetric tool for the diagnosis of fetal alcohol syndrome in infants," *IEEE Trans. Med. Imag.*, vol. 28, no. 9, pp. 1448–1458, Sep. 2009.
- [129] T. Mutsvangwa, E. M. Meintjes, D. L. Viljoen, and T. S. Douglas, "Morphometric analysis and classification of the facial phenotype associated with fetal alcohol syndrome in 5- and 12-year-old children," *Amer. J. Med. Genet. A*, vol. 152, pp. 32–41, 2010.
- [130] M. Suttie *et al.*, "Facial dysmorphism across the fetal alcohol spectrum," *Pediatrics*, vol. 131, no. 3, pp. 779–788, 2013.
- [131] S. Fang et al., "Automated diagnosis of fetal alcohol syndrome using 3D facial image analysis," Orthodontics Craniofacial Res., vol. 11, no. 3, pp. 162–171, 2008
- [132] K. R. Warren, B. G. Hewitt, and J. D. Thomas, "Fetal alcohol spectrum disorders: Research challenges and opportunities," *Alcohol Res. Health*, vol. 34, no. 1, pp. 4–15, 2011.
- [133] S. Boehringer *et al.*, "Syndrome identification based on 2D analysis software," *Eur. J. Human Genet.*, vol. 14, no. 10, pp. 1082–1089, 2006.
- [134] S. Boehringer *et al.*, "Automated syndrome detection in a set of clinical facial photographs," *Amer. J. Med. Genet. A*, vol. 155, no. 9, pp. 2161–2169, 2011.
- [135] M. Suttie *et al.*, "Facial dysmorphism across the fetal alcohol spectrum," *Pediatrics*, vol. 131, no. 3, pp. 779–788, 2013.
- [136] H. S. Loos, D. Wieczorek, R. P. Wrtz, C. von der Malsburg, B. Horsthemke, "Computer-based recognition of dysmorphic faces," *Eur. J. Human Genet.*, vol. 11, no. 8, pp. 555–560, 2003.
- [137] P. Hammond et al., "Discriminating power of localized threedimensional facial morphology," Amer. J. Human Genet., vol. 77, no. 6, pp. 999–1010, 2005.
- [138] D. Da Fonseca, M. Adida, R. Belzeaux, and J. M. Azorin, "Attention deficit hyperactivity disorder and/or bipolar disorder?," *Encephale*, vol. 40, pp. 23–26, 2014.
- [139] M. G. Sim, G. Hulse, and E. Khong, "When the child with ADHD grows up," Australian Family Physician, vol. 33, no. 8, pp. 615–618, 2004.
- [140] D. Delgado-Gomez *et al.*, "Microsoft kinect-based continuous performance test: An objective attention deficit hyperactivity disorder assessment," *J. Med. Internet Res.*, vol. 19, no. 3, pp. 1–8, 2017, Art. no. 79.
- [141] S. Jaiswal, M. Valstar, A. Gillott, and D. Daley, "Automatic detection of ADHD and ASD from expressive behaviour in RGBD data," in *Proc. IEEE Conf. Int. Conf. Autom. Face Gesture Recognit*, 2017, pp. 762–769.
- [142] F. Galgani, Y. Sun, P. L. Lanzi, and J. Leigh, "Automatic analysis of eye tracking data for medical diagnosis," in *Proc. IEEE Conf. Comput. Intell. Data Mining*, 2009, pp. 195–202.
- [143] M. Fried *et al.*, "ADHD subjects fail to suppress eye blinks and microsaccades while anticipating visual stimuli but recover with medication," *Vis. Res.*, vol. 101, pp. 62–72, 2014.

- [144] P. H. Tseng, I. G. Cameron, G. Pari, J. N. Reynolds, D. P Munoz, and L. Itti, "High-throughput classification of clinical populations from natural viewing eye movements," *J. Neurol.*, vol. 260, no. 1, pp 275–284, 2013.
- [145] S. Everling and B. Fischer, "The antisaccade: A review of basic research and clinical studies," *Neuropsychologia*, vol. 36, no. 9, pp. 885–899, 1998.
- [146] S. B. Hutton and U. Ettinger, "The antisaccade task as a research tool in psychopathology: A critical review," *Psychophysiology*, vol. 43, pp. 302–313, 2006.
- [147] N. J. Sasson, A. E. Pinkham, L. P Weittenhiller, D. J Faso, and C. Simpson, "Context effects on facial affect recognition in schizophrenia and autism: Behavioral and eye-tracking evidence," *Schizophrenia Bull.*, vol. 42, no. 3, pp. 675–683, 2016.
- [148] R. G. Ross, S. Heinlein, G. O. Zerbe, and A. Radant, "Saccadic eye movement task identifies cognitive deficits in children with schizophrenia, but not in unaffected child relatives," *J. Child Psychol. Psychiatry*, vol. 46, no. 12, pp. 1354–1362, 2005.
- [149] M. Asgharpour, M. Tehrani-Doost, M. Ahmadi and H. Moshki, "Visual attention to emotional face in schizophrenia: An eye tracking study," *Iranian J. Psychiatry*, vol. 10, no. 1, pp. 13–18, 2015.
- [150] P. Trillenberg *et al.*, "Visual and non-visual motion information processing during pursuit eye tracking in schizophrenia and bipolar disorder," *Eur. Arch. Psychiatry Clin. Neurosci.*, vol. 267, no. 3, pp. 225–235, 2017.
- [151] T. Falck-Ytter, S. Blte, and G. Gredebck, "Eye tracking in early autism research," J. Neurodev. Disorders, vol. 5, no. 1, pp. 5–28, 2013.
- [152] J. Hashemi *et al.*, "Computer vision tools for low-cost and noninvasive measurement of autism-related behaviors in infants," *Autism Res. Treatment*, vol. 2014, 2014, Art. no. 935686.
- [153] R. A. Cooper, K. C. Plaisted-Grant, S. Baron-Cohen, and J. S. Simons, "Eye movements reveal a dissociation between memory encoding and retrieval in adults with autism," *Cognition*, vol. 159, pp. 127–138, 2016.
- [154] K. Horley, L. Williams, C. Gonsalvez, and E. Gordon, "Social phobics do not see eye to eye: A visual scanpath study of emotional expression processing," *J. Anxiety Disorder*, vol. 17, no. 1, pp. 33–44, 2003.
- [155] M. N. Benfatto, G. Ö. Seimyr, J. Ygge, T. Pansell, A. Rydberg, and C. Jacobson, "Screening for dyslexia using eye tracking during reading," *PLoS One*, vol. 11, no. 12, pp. 1–16, 2016, Art. no. e0165508.
- [156] S. M. McClintock, M. M. Husain, T. L. Greer, and C. M. Cullum, "Association between depression severity and neurocognitive function in major depressive disorder: A review and synthesis," *Neuropsychology*, vol. 24, no. 1, pp. 9–34, 2010.
- [157] A. M. L. Koorevaara *et al.*, "Five personality and depression diagnosis, severity and age of onset in older adults," *J. Affective Disorder*, vol. 151, no. 1, pp.178–185, 2013.
- [158] J. Kohlinga, J. C. Ehrenthala, K. N. Levyb, H. Schauenburga, and U. Dingera, "Quality and severity of depression in borderline personality disorder: A systematic review and meta-analysis," *Clin. Psychol. Rev.*, vol. 37, pp. 13–25, 2015.
- [159] M. Prendergast, "Understanding depression," Melbourne Vic, Australia: Penguin Group, 2006.
- [160] V. Bevilacqua, D. D'Ambruoso, G. Mandolino, and M. Suma, "A new tool to support diagnosis of neurological disorders by means of facial expressions," in *Proc. IEEE Int. Workshop Med. Meas. Appl.*, 2011, pp. 544–549.
- [161] J. F. Cohn et al., "Detecting depression from facial actions and vocal prosody," in Proc. Conf. Affective Comput. Intell. Interact. Workshops, 2009, pp. 1–7.
- [162] J. Joshi, R. Goecke, G. Parker, and M. Breakspear, "Can body expressions contribute to automatic depression analysis?" in *Proc. 10th IEEE Int. Conf. Workshops Automat. Face Gesture Recognit.*, 2013, pp. 1–7.
- [163] P. Wang, C. Kohler, and R. Verma, "Estimating cluster overlap on manifolds and its application to neuropsychiatric disorders," in *Proc. IEEE Conf. Comput. Vis. Pattern Recognit.*, 2007, pp. 1–6.
- [164] B. L. Jian, C. L. Chen, W. L. Chu, and M. W. Huang, "The facial expression of schizophrenic patients applied with infrared thermal facial image sequence," *BMC Psychiatry*, vol. 17, pp. 1–7, 2017, Art. no. 229.
- [165] V. Engert *et al.*, "Exploring the use of thermal infrared imaging in human stress research," *PLoS One*, vol. 9, no. 3, pp. 1–11, 2014, Art. no. e90782.
- [166] M. Anbar, B. M. Gratt, and D. Hong, "Thermology and facial telethermography, Part I: History and technical review," *Dentomaxillofacial Radiol.*, vol. 27, no. 2, pp. 61–67, 1998.
- [167] B. M. Gratt and M. Anbar, "Thermology and facial telethermography: Part II: Current and future clinical applications in dentistry," *Dentomaxillofacial Radiol.*, vol. 27, no. 2, pp. 68–74, 1998.

- [168] S. D. Sikdar, A. Khandelwal, S. Ghom, R. Diwan, and F. M. Debta, "Thermography: A new diagnostic tool in dentistry," *J. Indian Acad. Oral. Med. Radiol.*, vol. 22, no. 4, pp. 206–210, 2010.
- [169] B. M. Gratt *et al.*, "Thermographic assessment of craniomandibular disorders: Diagnostic interpretation versus temperature measurement analysis," *J. Orofacial Pain*, vol. 8, pp. 278–288, 1994.
- [170] D. Canavan and B. M. Gratt, "Electronic thermography for the assessment of mild and moderate temporomandibular joint dysfunction," *Oral Surg. Oral Med. Oral Pathol. Oral Radiol. Endodontics*, vol. 9, no. 6, pp. 778–786, 1995.
- [171] S. B. McBeth and B. M. Gratt, "Thermographic assessment of temporomandibular disorders symptomology during orthodontic treatment," *Amer. J. Orthodontics Dentofacial Orthopedics*, vol. 109, pp. 481–488, 1996.
- [172] T. K. Kalili and B. M. Gratt, "Electronic thermography for the assessment of acute temporomandibular joint pain," *Compendium Continuing Educ. Dentistry*, vol. 17, pp. 979–983, 1996.
- [173] A. V. Dibai-Filho, A. C. Packer, A. C. Costa, and D. Rodrigues-Bigaton, "The chronicity of myogenous temporomandibular disorder changes the skin temperature over the anterior temporalis muscle," *J. Bodywork Movement Therapies*, vol. 18, no. 3, pp. 430–434, 2014.
- [174] A. V. Dibai Filho, A. C. Packer, A. C. Costa, and D. Rodrigues-Bigaton, "Accuracy of infrared thermography of the masticatory muscles for the diagnosis of myogenous temporomandibular disorder," *J. Manipulative Physiol. Therapeutics*, vol. 36, no. 4, pp. 245–252, 2013.
- [175] D. Rodrigues-Bigaton *et al.*, "Accuracy of two forms of infrared image analysis of the masticatory muscles in the diagnosis of myogenous temporomandibular disorder," *J. Bodywork Movement Therapies*, vol. 18, no. 1, pp. 49–55, 2014.
- [176] H. Fikackova and E. Ekberg, "Can infrared thermography be a diagnostic tool for arthralgia of the temporomandibular joint?," *Oral Surg. Oral Med. Oral Pathol. Oral Radiol. Endodontics*, vol. 98, no. 6, pp. 643–650, 2004.
- [177] A. V. Dibai-Filho *et al.*, "Women with more severe degrees of temporomandibular disorder exhibit an increase in temperature over the temporomandibular joint," *Saudi Dental J.*, vol. 27, no. 1, pp. 44–49, 2015.
- [178] D. S. Haddad, M. L. Brioschi, R. Vardasca, M. Weber, E. M. Crosato, E. S. Arita, "Thermographic characterization of masticatory muscle regions in volunteers with and without myogenous temporomandibular disorder: Preliminary results," *Dentomaxillofac Radiol.*, vol. 43, no. 8, pp. 1–7, 2014.
- [179] K. Woźniak, L. Szyszka-Sommerfeld, G. Trybek, and D. Piątkowska, "Assessment of the sensitivity, specificity, and accuracy of thermography in identifying patients with TMD," *Med. Sci. Monit.*, vol. 21, pp. 1485–1493, 2015.
- [180] B. Gencturk, V. V. Nabiyev, A. Ustubioglu, and S. Ketenci, "Automated pre-diagnosis of acromegaly disease using local binary patterns and its variants," in *Proc. 36th Int. Conf. Telecommun. Signal Process.*, 2013, pp. 817–821.
- [181] R. P. Kosilek *et al.*, "Diagnostic use of facial image analysis software in endocrine and genetic disorders: Review, current results and future perspectives," *Eur. J. Endocrinol.*, vol. 173, no. 4, pp. 39–44, 2015.
- [182] R. P. Kosilek *et al.*, "Automatic face classification of Cushing's syndrome in women—A novel screening approach," *Exp. Clin. Endocrinol. Diabetes*, vol. 121, no. 9, pp. 561–564, 2013.
- [183] S. Tominaga and T. Tanaka, "3-dimensional analysis of nystagmus using video and image processing," in *Proc. SICE Annu. Conf.*, 2010, pp. 89–91.
- [184] M. Azri, S. Young, H. Lin, C. Tan, and Z. Yang, "Diagnosis of ocular myasthenia gravis by means of tracking eye parameters." in *Proc. Int. Conf. IEEE Eng. Med. Biol. Soc.*, 2014, vol. 36, pp. 1460–1464.
- [185] S. R. Shih, H. Y. Li, Y. L. Hsiao, and T. C. Chang, "The application of temperature measurement of the eyes by digital infrared thermal imaging as a prognostic factor of methylprednisolone pulse therapy for graves' ophthalmopathy," *Acta Ophthalmol.*, vol. 88, no. 5, pp. 154–159, 2010.
- [186] J. W. Ryu, J. S. Paik, H. S. Hwang and Suk Woo Yang, "Diagnostic significance and usefulness in digital infrared thermal imaging (DITI) of patients with nonspecific orbital inflammation," *J. Korean Ophthalmol Soc.*, vol. 53, no. 12, pp. 1732–1736, 2012.
- [187] A. Nozawa and M. Tacano, "Correlation analysis on alpha attenuation and nasal skin temperature," J. Stat. Mech., Theory Exp., pp. 1–10, 2009, Art. no. P01007.

- [188] B. R. Nhan and T. Chau, "Classifying affective states using thermal infrared imaging of the human face," IEEE Trans. Biomed. Eng., vol. 57, no. 4, pp. 979-987, Apr. 2010.
- [189] C. Y. Hung et al., "Correlation between TCM inspection of facial color and pulmonary function of bronchial asthma," in Proc. IEEE Int. Conf. Bioinformat. Biomed. Workshops, 2010, pp. 754-757.
- [190] M. Liu and Z. H. Guo, "Hepatitis diagnosis using facial color image," Medical Biometrics. ICMB 2008 (vol. 4901 of Lecture Notes Computer Science). New York, NY, USA: Springer, 2007, pp. 160-167.
- [191] X.Wang, B. Zhang, Z. Guo, and D. Zhang," Facial image medical analysis system using quantitative chromatic feature," Expert Syst. Appl., vol. 40, no. 9, pp. 3738-3746, 2013.
- [192] J. A. Kors and A. L. Hoffmann, "Induction of decision rules that fulfill user-specified performance requirements," Pattern Recognit. Lett. vol. 18, pp. 1187-1195, 1997.
- [193] H. Muller et al., "A review of content-based image retrieval systems in medical applications-clinical benefits and future directions," Int. J. Med. Informat., vol. 73, no. 1, pp. 1-23, 2004.

- [194] J. H. Van Bemmel et al., "Databases for knowledge discovery: Examples from biomedicine and health care," Int. J. Med. Informat., vol. 75, no. 3/4, pp. 257-267, 2006.
- [195] B. Sadan, "Patient data confidentiality and patient rights," Int. J. Med. *Informat.*, vol. 62, no. 1, pp. 41–49, 2001. [196] Z. Liu, H. Wang, and Q. Li, "Tongue tumor detection in medical hyper-
- spectral images," Sensors, vol. 12, pp. 162-174, 2012.
- [197] M. Denstedt et al., "Wavelet based feature extraction and visualization in hyperspectral tissue characterization," Biomed. Opt. Expess, vol. 5, no. 12, pp. 4260-4280, 2014.
- [198] M. Uzair, A. Mahmood, and A. Mian, "Hyperspectral face recognition with spatiospectral information fusion and PLS regression," IEEE Trans. Image Process., vol. 24, no. 3, pp. 1127-1137, Mar. 2015.
- [199] Q. Li, "Hyperspectral imaging technology used in tongue diagnosis," in Recent Advances Theories Practice Chinese Medicine. Rijeka, Croatia: InTech, 2012, vol. 5.